Author's response to reviews

Title: Adherence to Highly Active Antiretroviral Therapy and Associated Factors among Children at Gondar University Hospital and Gondar Poly Clinic, North West Ethiopia.

Authors:

Berihun Assefa Dachew (berihunassefa21@gmail.com)
Tadis Brhane Tesfahunegn (taddis2007@gmail.com)
Anteneh Messele Birhanu (antenehmessele33@gmail.com)

Version: 2 Date: 9 October 2013

Author's response to reviews:

Dear Editors, Thank you very much for your constructive comments. The following points are changed in the revised version of the manuscript.
1. Parental Consent was included in the methodology part under ethical consideration section.
2. The grammatical and language edition was made starting from the title till the conclusion of the manuscript.

We thank you!

With Gratitude,